Author’s response to reviews

Title: Pathological profiles and clinical management challenges of breast cancer emerging in young women in Indonesia: a hospital-based study

Authors:

Sumadi Lukman Anwar (sl.anwar@ugm.ac.id)

Clarista Adelia Raharjo (clarista.ardelia@gmail.com)

Rahma Herviastuti (rahmahanggia@gmail.com)

Ery Kus Dwianingsih (ery_malueka@ugm.ac.id)

Didik Setyoheriyanto (didik_setyoheriyanto@mail.ugm.ac.id)

Widya Surya Avanti (w33ya@yahoo.com)

Lina Choridah (choridahlina@yahoo.com)

Wirsma Arif Harahap (wirsma_arif@yahoo.com)

Darwito Darwito (peraboi.yogyakarta@yahoo.co.id)

Teguh Aryandono (teguharyandono@yahoo.com)

Wahyu Wulaningsih (w.wulaningsih@ucl.ac.uk)

Version: 4 Date: 21 Jan 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for considering our manuscript BMWH-D-18-00388R3: “Pathological profiles and clinical management challenges of breast cancer emerging in young women in Indonesia: a hospital-based study” for publication. We are very pleased about the overall positive evaluation of our manuscript ("potentially acceptable for publication"). We appreciate all comments raised by the editorial team which provide us an opportunity to improve our manuscript. We thank for the offer to resubmit our revised version addressing all comments and suggestions. A point by point-reply describing all changes made in detail is included. The changes in the revised text are highlighted in blue.
Corrections to Technical comments:

1. Please clearly state whether the consent obtained from the participants was written or verbal

Done as requested, please see Page 14 line 13, our study participants have delivered written informed consent using procedures and forms that have been approved by the Ethical Committee board.

2. Consent for publication

We analyzed patient’s data anonymously in which image and personal data were not shown. Therefore, consent for publication is not applicable. Please see Page 14 line 15.

3. Formatting, Legends into Figure legends, and Conclusion into Conclusions

Done as suggested.

4. Figure citation

Table 3 has been mentioned at the Page 8 line 18.

5. Clean manuscript

Done as suggested